Author's response to reviews

Title: 47,XXY/48,XXYY/49,XXXXY mosaic with hydrocephaly: case report and literature review.

Authors:

  Jesus E Duenas (jedanet@yahoo.com)
  Maribel Aguilar (maribela2@excite.com)
  Eliakym Arambula (eliakymarambula@hotmail.com)
  Juliana B Valenzuela (juliana_70@yahoo.com)
  Angelina Vega (angyqfb@hotmail.com)
  Julio Granados (julgrate@yahoo.com)
  Rosalio Ramos (ramospayan@yahoo.com.mx)

Version: 4 Date: 17 August 2007

Author's response to reviews: see over
Dear Claire:
Enclosed please find the corrections and modifications on the about manuscript; I hope you will find it now suitable for publication.

Thank you very much for the effort you put on this matter.

Sincerely yours,

Rosalio Ramos Payan
FCQB-UAS
REVIEWER 1: ELISSA PALMER

ABSTRACT:
In line 3: the misspelled word “polisomy” was changed by “polysomy”.

BACKGROUND:
In line 4: the incorrect word “affection” was changed by “affliction”.
In line 6: the misspelled word “polisomy” was changed by “polysomy”.
In line 10: the phrase “Despite of being” was changed by “Despite being”.
In line 14: An “and” was inserted between “rare mosaic,” and “to our knowledge”.

CASE PRESENTATION:
In line 2: the incorrect sentence “He was born from young unrelated parents by cesarean at week 39 because cardiac pulse alteration and development stop in the last three weeks of gestation were observed” was changed by “He was born from young unrelated parents and birth by cesarean at week 39 due to fetal arrhythmias and intrauterine growth retardation”.
In line 4: the phrase “Apgar score 8/9 at 1 minute” was changed by “Apgar score of 8 at 1 minute and 9 at 5 minutes.
In line 5: the incorrect phrase “physical exploration” was changed by “clinical examination”.
In line 9: An “and” was inserted between “hypospadia” and “middle scrotal”.
In line 12: the incorrect word “conduct” was changed by “vessel”.
In line 13: the word “give” was changed by “gave”.
In line 15: the misspelled word “confirm” was changed by “confirmed”.

DISSCUSION:
In line 7: the misspelled word “determinate” was changed by “determinated”.
In line 12: the misspelled word “determinate” was changed by “determinated”.
In line 13: the misspelled word “report” was changed by “reported”.
In line 14: the incorrect phrase “it should be discarded” was changed by “it should be considered”.
REVIEWER 2: GUILLERMO ROMERO
In the ABSTRACT (line 3) and BACKGROUD (line 4) sections, the misspelled word “polisomy” was changed by “polysomy” instead of the suggested word “polysomies”.
ADDITIONALLY:
The corresponding author address was changed to Spanish.
In line 7 of the DISCUSSION section, the indication “(figure 1)” was inserted after “FISH analysis”.
In line 11 of the DISCUSSION section, the indication “(figure 2)” was inserted after “non-communicating hydrocephalus”.